Author’s response to reviews

Title: A successfully treated case of herpes simplex encephalitis complicated by subarachnoid bleeding: a case report

Authors:

Yasuyo Tonomura (tyasuyo@naramed-u.ac.jp)
Hiroshi Kataoka (hk55@naramed-u.ac.jp)
Noritaka Yata (nyata@naramed-u.ac.jp)
Makoto Kawahara (kawamm@naramed-u.ac.jp)
Kazuo Okuchi (kokuchi@naramed-u.ac.jp)
Satoshi Ueno (sueno@naramed-u.ac.jp)

Version: 5  Date: 20 July 2010

Author’s response to reviews: see over
July 20, 2010

Manuscript ID MS: 1709494568371904
"A successfully treated case of herpes simplex encephalitis complicated by subarachnoid bleeding: a case report"

Dear The Independent Editorial Production Team of Journal of Medical Case Reports,

We are returning herewith the above manuscript revised according to your letter dated on July 19th, 2010. We revised our manuscript.

Thank you for your corporations.

Best regards,

Hiroshi Kataoka, MD.PhD
Department of Neurology
Nara Medical University
Email: hk55@naramed-u.ac.jp

Major revisions
Patient information: please provide the ethnicity of the patient in the case presentation section of the abstract and the main text of the manuscript. As suggested, we added them.